REVIEW HISTORY

Paper: "An Investigation of Interpersonal Problem Solving in University Students in Terms of Personality Traits, Resilience and Hope"

Corresponding Author: Ahmet Alkal

Email: aalkal@bingol.edu.tr

Doi: 10.19044/ejes.v7no1a2

Peer review:

Reviewer 1: Blinded

Reviewer 2: Blinded

Published: 31.03.2020

